Author's response to reviews

Title: Reports of evidence planting by police among a community-based sample of injection drug users in Bangkok, Thailand

Authors:

Nadia Fairbairn (n.fairbairn@gmail.com)
Karyn Kaplan (karyn.kaplan@gmail.com)
Kanna Hayashi (kanna.hayashi@gmail.com)
Calvin Lai (clai@cfenet.ubc.ca)
Evan Wood (uhri@cfenet.ubc.ca)
Thomas Kerr (uhri@cfenet.ubc.ca)

Version: 6 Date: 26 August 2009

Author's response to reviews: see over
August 25, 2009

Natalie Pafitis MSc
Senior Scientific Editor
BMC-series Journals
BioMed Central

Dear Natalie Pafitis and Colleagues:

Re: MS: 2082282301261212 – “EVIDENCE PLANTING BY POLICE AMONG A COMMUNITY-BASED SAMPLE OF INJECTION DRUG USERS IN BANGKOK, THAILAND”

Thank you for your email dated August 25, 2009. We are delighted that our manuscript will be published in BMC International Health and Human Rights. We have checked the manuscript over carefully and made a few typographical changes. An updated version of the manuscript has been prepared, and details of the revisions are provided here. If there are any other changes you require, please do not hesitate to ask. Thank you again for your offer of publication.

Yours sincerely,

[Signature]
Thomas Kerr, PhD
uhri@cfenet.ubc.ca
Proofing

1. We have revised one sentence in the second paragraph on page 7 (Discussion and Conclusions)
   
   “Though causal relationships can not be inferred in the present study…”

2. We have placed the web-links for references 16, 17, 18, and 20 on a new line.

3. We have removed the typographical error in Reference 11 by deleting the date “(1982)”.

4. In the title of Table 1, we have removed the capitalization from the word “injection”.